Author's response to reviews

Title: Measurement properties of the Western Ontario Shoulder Instability Index in Dutch patients with shoulder instability

Authors:

Just A. van der Linde (javdlinde@gmail.com)
W. Jaap Willems (w.j.willems@xs4all.nl)
Derk A. van Kampen (derkvankampen@gmail.com)
Loes W.A.H. van Beers (L.W.A.H.vanBeers@olvg.nl)
Derek F.P. van Deurzen (D.F.P.vanDeurzen@olvg.nl)
Caroline B. Terwee (cb.terwee@vumc.nl)

Version: 7
Date: 19 May 2014

Author's response to reviews:

Dear sir, madam,

We have revised our manuscript regarding the following aspects

The name of ethics committee:
- Please update your ethics statement to include the name of the ethics committee that approved your study.

Requesting consent statement:
- Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

We added to line 131:

‘The local ethics committee (METC from the Onze Lieve Vrouwe Gasthuis) approved the study and written informed consent was obtained from all patients.’

Sincerely,

Just van der Linde